Author's response to reviews

Title: The variability and predictors of quality in scaled up AIDS care services: an observational study in Brazil.

Authors:

Maria IB Nemes (mibnemes@usp.br)
Regina Melchior (reginamelchior@gmail.com)
Cáritas R Basso (cr_basso@terra.com.br)
Elen RL Castanheira (elen@fmb.unesp.br)
Maria TSSB Alves (seabra@elo.com.br)
Shaun Conway (shaun@trendresponse.com)

Version: 2 Date: 12 June 2008

Author's response to reviews:

Dear Editor

Thank you for your attention.

According to your suggestions we are submitting a revision of the paper “The variability and predictors of quality in scaled up AIDS care services: an observational study in Brazil”.

We have provided:

- A copy of the full version of the questionnaire in Portuguese language. We also uploaded a short version in English.
- A revision of the abstract
- More information on responders versus non-responders

Yours sincerely,

Elen Rose Lodeiro Castanheira on behalf of all authors.